Author’s response to reviews

Title: Understanding primary care-oncology relationships within a changing healthcare environment

Authors:

Jennifer Tsui (jk1341@cinj.rutgers.edu)
Jenna Howard (jenna.howard@rutgers.edu)
Denalee O’Malley (omalledm@rwjms.rutgers.edu)
William L. Miller (william.miller@lvhn.org)
Shawna V. Hudson (shawna.hudson@rutgers.edu)
Ellen B. Rubinstein (ellen.rubinstein@aya.yale.edu)
Jeanne M. Ferrante (jeanne.ferrante@rutgers.edu)
Alicja Bator (alicjab@rutgers.edu)
Benjamin F. Crabtree (benjamin.crabtree@rutgers.edu)

Version: 2 Date: 18 Nov 2019

Author’s response to reviews:

We thank the reviewers/editors for the thoughtful review of our revised manuscript. Here are the edits we made in response to comments:

1. Consent to participate – clarify if written or verbal
   In your “ethical approval and consent to participate” section, please confirm whether informed consent, obtained from all patients, was written or verbal and clearly state this in your manuscript. If verbal, please state the reason and whether the ethics committee approved this procedure.

   We clarified "Written informed consent" was obtained from all recorded interview participants. This was edited in the methods section of the manuscript as well as the “ethical approval and consent to participate” section.

2. Consent for publication
   In your ‘Consent for publication’ section of your declarations, please confirm whether all participants gave written consent for their personal or clinical details along with any identifying images to be published in this study. Consent for publication must be in writing, and cannot be
waived if it is necessary. Alternatively, if you believe all identifying information has been appropriately anonymised (e.g. by grouping ages into age ranges) and there are no identifying images used within the manuscript, then consent to publish is not applicable to your manuscript - please write ‘Not Applicable’ in this section if this is the case.

We have indicated that this is not applicable to our manuscript.

3. Role of Funders
Please note the role(s) of the funding body (bodies) in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript, which should be declared in the Funding section.

We have clarified the funding bodies have not had a role.

4. Remove supplementary file
Please remove the supplementary “Cover Letter.docx”, as these files are not required to be attached to the manuscript at this stage of the editorial process.

We removed the cover letter.

5. Clean manuscript
At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

We have added a clean version of the manuscript.